\begin{thebibliography}{10}
\providecommand{\url}[1]{\texttt{#1}}
\providecommand{\urlprefix}{\\URL: }
\providecommand{\bibAnnoteFile}[1]{%
  \IfFileExists{#1}{\begin{quotation}\noindent\textsc{Key:} #1\\
  \textsc{Annotation:}\ \input{#1}\end{quotation}}{}}
\providecommand{\bibAnnote}[2]{%
\begin{quotation}\noindent\textsc{Annotation:}\ #2\end{quotation}}

\bibitem{Geesaman:2015fha}
A.~Aprahamian \emph{et~al.}, \emph{{Reaching for the Horizon: The 2015 Long
  Range Plan for Nuclear Science}}, Nuclear Science Advisory Committee  (2015),
  \urlprefix\url{{http://science.energy.gov/~/media/np/nsac/pdf/2015LRP/2015_LRPNS_091815.pdf}}
\bibAnnoteFile{Geesaman:2015fha}

\bibitem{walker:CISA20}
{Walker, P. and Dugger, M.}, \emph{Isolation of the phi-meson from $e p
  \rightarrow e p K^+ K^-$ events} (presentation at the College of Integrative
  Sciences and Arts Virtual Student Research Day, Dec. 2020, Polytechnic
  Campus, Arizona State),
  \urlprefix\url{http://meson.hldsite.com/presentations/CISAFall20/PatrickWalker_CISA_Fall_2020.pdf}
\bibAnnoteFile{walker:CISA20}

\bibitem{walker:PhysSymposium}
{Walker, P.}, \emph{{Search for new states that decay $K^* K"$}} (presentation
  at the 17th Annual Undergraduate Physics Symposium, May 2021, Tempe Campus,
  Arizona State University), \urlprefix\url{https://youtu.be/Buu2YDIrUK4}
\bibAnnoteFile{walker:PhysSymposium}

\bibitem{osar:CISA20}
{Osar R. and Dugger M.}, \emph{Event generator for $\gamma p \rightarrow p K K$
  events} (presentation at the College of Integrative Sciences and Arts Virtual
  Student Research Day, Dec. 2020, Polytechnic Campus, Arizona State),
  \urlprefix\url{http://meson.hldsite.com/presentations/CISAFall20/RebeccaOsar_CISA_Fall_2020.pdf}
\bibAnnoteFile{osar:CISA20}

\bibitem{osar:PhysSymposium}
{Osar, R.}, \emph{{Search for high-mass resonances decaying to $K \Lambda^*"$}}
  (presentation at the 17th Annual Undergraduate Physics Symposium, May 2021,
  Tempe Campus, Arizona State University),
  \urlprefix\url{https://youtu.be/Q2YhBCv1riE}
\bibAnnoteFile{osar:PhysSymposium}

\bibitem{lamagna:PhysSymposium}
{Lamagna, E.}, \emph{{On the road to constructing events that decay $\phi
  \pi^0$ using electroproduction with a proton target}} (presentation at the
  17th Annual Undergraduate Physics Symposium, May 2021, Tempe Campus, Arizona
  State University), \urlprefix\url{https://youtu.be/3AZP9fjStuc}
\bibAnnoteFile{lamagna:PhysSymposium}

\bibitem{costelle:PhysSymposium}
{Costelle, A.}, \emph{{Construction of Event Generators for
  Strangeness-Containing Final States}} (presentation at the 17th Annual
  Undergraduate Physics Symposium, May 2021, Tempe Campus, Arizona State
  University), \urlprefix\url{https://youtu.be/tKCB4t_G2yk}
\bibAnnoteFile{costelle:PhysSymposium}

\bibitem{momo:poster1}
{Mohamed, M. and Dugger, M.}, \emph{Simulations for energy depositions in a
  phantom cell of water for different types of therapy beams} (presentation at
  the College of Integrative Sciences and Arts Student Research Day, Dec. 2019,
  Polytechnic Campus, Arizona State),
  \urlprefix\url{http://www.public.asu.edu/~dugger/student2019/PosterMoMoV2.pdf}
\bibAnnoteFile{momo:poster1}

\bibitem{momo:poster2}
{Mohamed, M. and Dugger, M.}, \emph{Particle identification study using
  decision trees} (presentation at the College of Integrative Sciences and Arts
  Student Research Day, May 2020, Polytechnic Campus, Arizona State),
  \urlprefix\url{http://meson.hldsite.com/presentations/Poster2020/PosterMoMoSp20Final.pdf}
\bibAnnoteFile{momo:poster2}

\bibitem{kevin:poster}
{Scheuer, K. and Cole, S. and Dugger, M.}, \emph{Evaluation of machine learning
  methods on simulated tracks to improve K pi identification in GlueX}
  (presentation at the College of Integrative Sciences and Arts Student
  Research Day, May 2020, Polytechnic Campus, Arizona State),
  \urlprefix\url{http://meson.hldsite.com/presentations/Poster2020/PresentationKevin2020.pdf}
\bibAnnoteFile{kevin:poster}

\bibitem{gluexPac30}
G.~Collaboration, \emph{{Mapping the Spectrum of Light Quark Mesons and Gluonic
  Excitations with Linearly Polarized Photons}}, {Jefferson Lab Experiment
  Proposal PR-06-102}, GlueX Collaboration, Jefferson Lab (2006),
  \urlprefix\url{{https://halldweb.jlab.org/DocDB/0012/001226/001/pac30.pdf}}
\bibAnnoteFile{gluexPac30}

\bibitem{gx2year}
C.~Meyers, \emph{{Potential GlueX Physics During the First Two Years of 12 GeV
  Running}}, {Report}, GlueX Collaboration, Jefferson Lab (2006),
  \urlprefix\url{{https://halldweb.jlab.org/DocDB/0006/000621/002/gluex_pac.pdf}}
\bibAnnoteFile{gx2year}

\bibitem{Adam:2004fq}
I.~Adam \emph{et~al.}, \emph{{The DIRC particle identification system for the
  BaBar experiment}}, Nucl. Instrum. Meth. \textbf{A538} (2005), pp. 281--357
\bibAnnoteFile{Adam:2004fq}

\bibitem{Zyla:2020zbs}
P.~Zyla \emph{et~al.}, \emph{{Review of Particle Physics}}, PTEP \textbf{2020}
  (2020)~(8), p. 083C01
\bibAnnoteFile{Zyla:2020zbs}

\bibitem{Capstick:1986bm}
S.~Capstick and N.~Isgur, \emph{{Baryons in a relativized quark model with
  chromodynamics}}, Phys. Rev. \textbf{D34} (1986), p. 2809, [AIP Conf.
  Proc.132,267(1985)]
\bibAnnoteFile{Capstick:1986bm}

\bibitem{Crede:2017kyh}
V.~Crede and S.~Capstick, \emph{{Bound States of $N$'s and $\Xi$'s}}, in
  \emph{{Workshop on Excited Hyperons in QCD Thermodynamics at Freeze-Out
  (YSTAR2016) Mini-Proceedings}} (2017), pp. 40--45,
  \urlprefix\url{http://inspirehep.net/record/1511541/files/1510585_40-45.pdf}
\bibAnnoteFile{Crede:2017kyh}

\bibitem{Dudek:2013yja}
J.~J. Dudek, R.~G. Edwards, P.~Guo and C.~E. Thomas, \emph{{Toward the excited
  isoscalar meson spectrum from lattice QCD}}, Phys. Rev. \textbf{D88}
  (2013)~(9), p. 094505
\bibAnnoteFile{Dudek:2013yja}

\bibitem{Chung:2002fz}
S.~U. Chung, E.~Klempt and J.~G. Korner, \emph{{SU(3) classification of
  $p-$wave $\eta \pi$ and $\eta^\prime \pi$ systems}}, Eur. Phys. J.
  \textbf{A15} (2002), pp. 539--542
\bibAnnoteFile{Chung:2002fz}

\bibitem{Iddir:2000yb}
F.~Iddir and A.~S. Safir, \emph{{The Decay of the observed $J^{PC} =
  1^{-+}(1400)$ and $J^{PC}=1^{-+}(1600)$ hybrid candidates}}, Phys. Lett.
  \textbf{B507} (2001), pp. 183--192
\bibAnnoteFile{Iddir:2000yb}

\bibitem{Close:1987aw}
F.~E. Close and H.~J. Lipkin, \emph{{New Experimental Evidence for Four Quark
  Exotics:The Serpukhov $\phi \pi$ Resonance and the Gams $\eta \pi$
  Enhancement}}, Phys. Lett. \textbf{B196} (1987), p. 245
\bibAnnoteFile{Close:1987aw}

\bibitem{Intensity_V}
V.~Mathieu, M.~Albaladejo, C.~FernÃ¡ndez-RamÃ­rez, A.~Jackura, M.~Mikhasenko,
  A.~Pilloni and A.~Szczepaniak, \emph{Moments of angular distribution and beam
  asymmetries in eta pi0 photoproduction at GlueX}, Physical Review D
  \textbf{100} (2019)~(5),
  \urlprefix\url{http://dx.doi.org/10.1103/PhysRevD.100.054017}
\bibAnnoteFile{Intensity_V}

\bibitem{CLAS:2021udy}
N.~Zachariou \emph{et~al.}, \emph{{Double polarisation observable $G$ for
  single pion photoproduction from the proton}}, Phys. Lett. B \textbf{817}
  (2021), p. 136304
\bibAnnoteFile{CLAS:2021udy}

\bibitem{CLAS:2021osv}
U.~Shrestha \emph{et~al.}, \emph{{Differential cross sections for
  $\Lambda(1520)$ using photoproduction at CLAS}}, Phys. Rev. C \textbf{103}
  (2021)~(2), p. 025206
\bibAnnoteFile{CLAS:2021osv}

\bibitem{Hayward:2021psm}
T.~B. Hayward \emph{et~al.}, \emph{{Observation of Beam Spin Asymmetries in the
  Process $ep\rightarrow{e}^{'}{\pi}^{+}{\pi}^{-}X$ with CLAS12}}, Phys. Rev.
  Lett. \textbf{126} (2021), p. 152501
\bibAnnoteFile{Hayward:2021psm}

\bibitem{CLAS:2020ngl}
M.~Carver \emph{et~al.}, \emph{{Photoproduction of the $f_2(1270)$ meson using
  the CLAS detector}}, Phys. Rev. Lett. \textbf{126} (2021)~(8), p. 082002
\bibAnnoteFile{CLAS:2020ngl}

\bibitem{PhysRevC.103.L022201}
S.~Adhikari \emph{et~al.}, \emph{Measurement of beam asymmetry for
  ${\ensuremath{\pi}}^{\ensuremath{-}}{\mathrm{\ensuremath{\Delta}}}^{++}$
  photoproduction on the proton at
  ${E}_{\ensuremath{\gamma}}=8.5\phantom{\rule{0.16em}{0ex}}\mathrm{GeV}$},
  Phys. Rev. C \textbf{103} (2021), p. L022201,
  \urlprefix\url{https://link.aps.org/doi/10.1103/PhysRevC.103.L022201}
\bibAnnoteFile{PhysRevC.103.L022201}

\bibitem{CLAS:2020vpr}
T.~Hu \emph{et~al.}, \emph{{Photoproduction of $\eta$ mesons off the proton for
  $1.2 < E_\gamma < 4.7$ GeV using CLAS at Jefferson Laboratory}}, Phys. Rev. C
  \textbf{102} (2020)~(6), p. 065203
\bibAnnoteFile{CLAS:2020vpr}

\bibitem{ADHIKARI2021164807}
S.~Adhikari \emph{et~al.}, \emph{The GlueX beamline and detector}, Nuclear
  Instruments and Methods in Physics Research Section A: Accelerators,
  Spectrometers, Detectors and Associated Equipment \textbf{987} (2021), p.
  164807,
  \urlprefix\url{https://www.sciencedirect.com/science/article/pii/S0168900220312043}
\bibAnnoteFile{ADHIKARI2021164807}

\bibitem{CLAS:2020rdz}
A.~Celentano \emph{et~al.}, \emph{{First measurement of direct photoproduction
  of the $a_2(1320)^0$ meson on the proton}}, Phys. Rev. C \textbf{102}
  (2020)~(3), p. 032201
\bibAnnoteFile{CLAS:2020rdz}

\bibitem{PhysRevC.101.065206}
S.~Adhikari \emph{et~al.}, \emph{Measurement of the photon beam asymmetry in
  $\vec{\gamma}p\ensuremath{\rightarrow}{K}^{+}{\mathrm{\ensuremath{\Sigma}}}^{0}$
  at ${E}_{\ensuremath{\gamma}}=8.5$ GeV}, Phys. Rev. C \textbf{101} (2020), p.
  065206, \urlprefix\url{https://link.aps.org/doi/10.1103/PhysRevC.101.065206}
\bibAnnoteFile{PhysRevC.101.065206}

\bibitem{CLAS:2020mom}
A.~Schmidt \emph{et~al.}, \emph{{Probing the core of the strong nuclear
  interaction}}, Nature \textbf{578} (2020)~(7796), pp. 540--544
\bibAnnoteFile{CLAS:2020mom}

\bibitem{PhysRevC.100.052201}
S.~Adhikari \emph{et~al.}, \emph{Beam asymmetry $\mathrm{\ensuremath{\Sigma}}$
  for the photoproduction of $\ensuremath{\eta}$ and
  ${\ensuremath{\eta}}^{\ensuremath{'}}$ mesons at
  ${E}_{\ensuremath{\gamma}}=8.8\phantom{\rule{0.28em}{0ex}}\mathrm{GeV}$},
  Phys. Rev. C \textbf{100} (2019), p. 052201,
  \urlprefix\url{https://link.aps.org/doi/10.1103/PhysRevC.100.052201}
\bibAnnoteFile{PhysRevC.100.052201}

\bibitem{PhysRevLett.123.072001}
A.~Ali \emph{et~al.}, \emph{First Measurement of Near-Threshold
  $J/\ensuremath{\psi}$ Exclusive Photoproduction off the Proton}, Phys. Rev.
  Lett. \textbf{123} (2019), p. 072001,
  \urlprefix\url{https://link.aps.org/doi/10.1103/PhysRevLett.123.072001}
\bibAnnoteFile{PhysRevLett.123.072001}

\bibitem{CLAS:2018mmb}
P.~Roy \emph{et~al.}, \emph{{First Measurements of the Double-Polarization
  Observables F , P , and H in \ensuremath{\omega} Photoproduction off
  Transversely Polarized Protons in the N* Resonance Region}}, Phys. Rev. Lett.
  \textbf{122} (2019)~(16), p. 162301
\bibAnnoteFile{CLAS:2018mmb}

\bibitem{PhysRevC.88.065203}
M.~Dugger \emph{et~al.}, \emph{Beam asymmetry $\ensuremath{\Sigma}$ for
  ${\ensuremath{\pi}}^{+}$ and ${\ensuremath{\pi}}^{0}$ photoproduction on the
  proton for photon energies from 1.102 to 1.862 GeV}, Phys. Rev. C \textbf{88}
  (2013), p. 065203,
  \urlprefix\url{https://link.aps.org/doi/10.1103/PhysRevC.88.065203}
\bibAnnoteFile{PhysRevC.88.065203}

\bibitem{talk:coleDNP19}
{Cole, S.}, \emph{{$K^* K$ in GlueX}} (presentation at Fall Meeting of the
  American Physical Society Division of Nuclear Physics, 2019),
  \urlprefix\url{http://www.public.asu.edu/~dugger/student2019/KStarK-DNP-10-14-19.pdf}
\bibAnnoteFile{talk:coleDNP19}

\bibitem{talk:sumnerDNP21}
{Sumner, B.}, \emph{{Search for excited $\Xi^*$ states and preliminary cross
  section for $\Xi^-(1530)$}} (presentation at the American Physical Society
  April Meeting, 2021),
  \urlprefix\url{http://meson.hldsite.com/presentations/Sumner/APSSumnerApril2021.pdf}
\bibAnnoteFile{talk:sumnerDNP21}

\bibitem{clasDatabase}
{CLAS Collaboration}, \emph{JLAB Experiment {CLAS} Database},
  \urlprefix\url{http://clasweb.jlab.org/physicsdb/intro.html}
\bibAnnoteFile{clasDatabase}

\bibitem{SAIDdatabase}
{Institute for Nuclear Studies Data Analysis Center}, \emph{{SAID Partial-Wave
  Analysis Facility}}, \urlprefix\url{http://gwdac.phys.gwu.edu/}
\bibAnnoteFile{SAIDdatabase}

\bibitem{evioUserGuide}
{Jefferson Lab Data Acquisition Group}, \emph{EVIO Users' Guide},
  \urlprefix\url{ftp://ftp.jlab.org/pub/coda/evio/2.0/evio_Users_Guide.pdf}
\bibAnnoteFile{evioUserGuide}

\end{thebibliography}

\vspace{-5pt}
\subsection{Past accomplishments and works in progress}
\vspace{-5pt}

\label{sec:past}

The ASU Meson Physics Group has been working on electro- and photo-produced reactions from the proton.
For the electro-production work we have primarily utilized undergraduates to isolate a variety of 
strangeness containing states. 
The graduate students work on GlueX photoproduction data to uncover $\Xi^\ast$ baryons, and 
meson states that decay $K^+ K^- \pi^0$. 
%to isolate resonant mesons that decay
%through $K^+ K^- \pi^0$ and to isolate excited $\Xi$ baryon states.

\begin{wrapfigure}[]{r}{0.4\textwidth}
\vspace{-15pt}
\begin{center}
\includegraphics[width=\linewidth]{osarSymp.png}
\end{center}
\vspace{-20pt}
\caption{Invariant mass of $K^+ \Lambda(1520)$.
\newline
\vspace{-8pt}
}
{\label{fig:osarSymp}}
\end{wrapfigure}
Due to the transient and unpredictable nature of including undergraduates in advanced scientific study, 
the direction of the undergraduate work is completely determined by opportunity and is not planned 
far in advance.   
As stated in the introduction, our undergraduates have investigated 
$e p \rightarrow e p K^* K$ (Patrick Walker~\cite{walker:CISA20,walker:PhysSymposium}), 
$e p \rightarrow e K \Lambda(1520)$ (Rebecca Osar~\cite{osar:CISA20,osar:PhysSymposium}),
$e p \rightarrow e p \phi \pi$ (Emily Lamanga~\cite{lamagna:PhysSymposium}) and
$\gamma p \rightarrow p \phi$ (Anna Costelle~\cite{costelle:PhysSymposium}). In addition to the 
work listed above, the undergraduates have participated in the investigation of 
machine learning algorithms (Mohamed Mohamed~\cite{momo:poster2}, Kevin Scheuer~\cite{kevin:poster}).
Some of the undergraduates have won local ASU awards for their work: 
Patrick Walker (ASU Department of Physics Research Award),
Anna Costelle (ASU Women in Physics Award) and
Mohamed Mohamed (CISA Undergraduate Research Award).


In order to keep the discussion of our undergraduate work to be of reasonable length, 
I only highlight the work 
performed by Ms. Osar towards obtaining cross section measurements for 
$e p \rightarrow e N^*$ 
where $N^* \rightarrow K \Lambda(1520)$.
Ms. Osar's analysis has progressed more than the other undergraduates in the group. She
has background subtracted $\Lambda(1520)$ signals from invariant mass $K^- p$ events
for each bin in invariant mass $K^+ \Lambda(1520)$ shown in Figure~\ref{fig:osarSymp}. The fit
shown on the figure (solid black line) has a Gaussian center at 2.299 GeV that coincides 
with the two-star $N(2300)$ 
resonance. Additionally, there is an interesting
feature near 3.4 GeV that should be explored.
Ms. Osar has recently completed work on a event generator for the 
reaction and we are in the process of determining the detector efficiency for the 
mass bins given in Figure~\ref{fig:osarSymp}.  


The GlueX photoproduction data on mesons that decay $K^\ast \bar{K}$ 
and the isolation of excited $\Xi$ baryon states forms the most important component of our research
and is discussed next.

%The graduate students work to isolate resonant mesons that decay
%through $K^+ K^- \pi^0$ and to isolate excited $\Xi$ baryon states.

\vspace{-10pt}
\subsubsection{Work on $K^\ast \bar{K}$ states}
\vspace{-5pt}

Former ASU graduate student Sebastian Cole studied mesons that decay by way of $K^\ast \bar{K}$. 
The invariant mass of $K^{-}\pi^0$ from $\gamma p \rightarrow pK^{+}K^{-}\gamma\gamma$ events is shown
in Figure~\ref{fig:kmStar} with
vertical red lines 
%at $0.8~\mathrm{GeV}$ and $1.0~\mathrm{GeV}$ 
indicating the selection of events with $K^{*-}(892)$ meson. 
Taking those events that are consistent with the $K^{*-}(892)$ meson, an invariant mass of $K^{*-}(892)K^+$
is formed (see Figure~\ref{fig:kmStarKp}). Very similar results are obtained for the $K^{*+}(892)K^-$ case where
$K^{*+}(892)$ is formed from $K^{+}\pi^0$. 

\vspace{15pt}
\begin{figure}[h]
\begin{minipage}[t][1.5in][t]{.5\linewidth}
 { \centering
  \captionsetup{margin=0.4cm}
  \includegraphics[width=0.9\linewidth]{hKStarMinus_new.pdf}
  \caption{Invariant mass of $K^{-}\pi^0$.
  %\vspace{-5pt}
  }
  \label{fig:kmStar}}
\end{minipage}%                                                                                       
\begin{minipage}[t][1.5in][t]{.5\linewidth}
 { \centering
  \captionsetup{margin=0.4cm}
  \includegraphics[width=0.9\linewidth]{hKStarMinus_KPlus_new.pdf}
  \caption{Invariant mass of $K^{*-}(892)K^+$. 
    %\vspace{-5pt} 
  }
  \label{fig:kmStarKp}}
\end{minipage}
\end{figure}
\vspace{25pt}



%XXXXXXXXXXXXXXXXXX

%A cross section $\sigma$ for a given 
%process is defined as the number of interaction per unit time per target 
%particle divided by the incident flux. With respect to particular 
%kinematic variables, the intensity for the production of meson states 
%can be defined using the differential cross section. The intensity is

%\begin{align}
%    I(\Omega,\Omega_H,\Phi) &\equiv \frac{d\sigma}{dt~dm_{K^*\bar{K}}~d\Omega~d\Omega_H~d\Phi},
%\end{align}

%\noindent where $\Omega$ is the direction of the $K^*$ in the helicity frame 
%of the system, $\Omega_H$ is the direction of the $K$ in the helicity frame of the photoproduced 
%meson resonance, $\Phi$ the direction of the photon beam polarization 
%with respect to the production plane, $m_{K^*\bar{K}}$ is the mass of the meson 
%resonance, and $t$ is the Mandelstam momentum transfer 

%Using the formalism introduced by V. Mathieu \cite{Intensity_V}, 
%which is expanded by M. Shepherd \cite{Intensity_S} and J. Stevens \cite{Intensity_J}, 
%the intensity can be defined in helicity space. 
The intensity utilizing a polarized photon beam can be written as~\cite{Intensity_V}
\begin{align}
    I(\Omega,\Omega_H,\Phi) &= I^0(\Omega,\Omega_H) - P_\gamma I^1(\Omega,\Omega_H)\cos(2\Phi) - P_\gamma I^2\sin(2\Phi),
    \label{equation:I_exp}
\end{align}
where $P_\gamma$ is the degree of the linear polarization of the incident photon beam. Each $I$ in Equation
\ref{equation:I_exp} can be defined in terms of decay amplitudes:

\begin{align}
    I^0(\Omega,\Omega_H) &= \frac{\kappa}{2} \sum_\lambda A_\lambda(\Omega,\Omega_H)A^\ast_\lambda(\Omega,\Omega_H)\\
    I^1(\Omega,\Omega_H) &= \frac{\kappa}{2} \sum_\lambda A_{-\lambda}(\Omega,\Omega_H)A^\ast_\lambda(\Omega,\Omega_H)\\
    I^2(\Omega,\Omega_H) &= i\frac{\kappa}{2} \sum_\lambda \lambda A_{-\lambda}(\Omega,\Omega_H)A^\ast_\lambda(\Omega,\Omega_H).
\end{align}
The amplitudes are then expanded in terms of partial waves
\begin{align}
    A_\lambda &= \sum_{i}\sum_{m}T^i_{\lambda,m}B,
    \label{equation:partial_wave}
\end{align}
where $T$ are the partial wave amplitudes and $B$  
%the decay amplitude.
%The amplitude defined in Equation \ref{equation:partial_wave} used to determine the 
%intensity can be defined in terms of two rotations on spin-space. Let $B$ be 
the matrix element associated with the two rotations of interest:
\begin{align}
    B &\equiv \langle j_1m_1j_2m_2|R^{(1)}R^{(2)}|Jm \rangle,
\end{align}
where the $j_1$ represents the quantum number for the orbital angular momentum from the decay of 
the resonance meson to the $K^\ast$ vector meson and the pseudoscalar $\bar{K}$ meson, with the $j_2$ quantum number
being associated with the spin of the $K^\ast$ vector meson. The rotations 
$R^{(1)}$ and $R^{(2)}$ act on $|j_1m_1\rangle$ and $|j_2m_2\rangle$, 
respectively. 
%The rotation angles $\theta$ (polar) $\phi$ (azimuth) in the vector meson helicity frame define the 
%direction of the $K^\ast$, with the rotation angles $\theta_H$ (polar), and $\phi_H$ (azimuth) of the $\pi$
%as measured in the helicity frame of the $K^\ast$. 

%For the vector-pseudoscalar decay of interest, rotations with respect to the angles 
%$\theta$, $\phi$, $\theta_H$, and $\phi_H$ are defined in the resonance and vector meson rest frames. 
%As a result, $R^{(1)}$ is the rotation in the resonance rest frame, $R_r(\theta,\phi)$, 
%and $R^{(2)}$ is the rotation in the vector meson rest frame, $R_v^{(2)}R_r^{(2)}$. 
%Therefore, the matrix element can be written as
%\begin{align}
%    B &= \langle j_1m_1j_2m_2| R_r^{(1)}R_v^{(2)}R_r^{(2)} |Jm \rangle.
%    \label{eq:decay_amp2}
%\end{align}
%\noindent If a complete set of spin projections with respect to $j_1$ and $j_2$ is introduced to 
%Equation \ref{eq:decay_amp2}, $B$ can be written in terms of Wigner D-functions.
%The Wigner D-functions are related to the spin-weighted spherical harmonics, operating to move 
%between frames of the sequential steps of a meson decay using rotations on the sp\
%in space. The Euler angles $(\alpha,\beta,\gamma)$ for which the Wigner D-function is defined
%\begin{align}
%    D^J_{m'm} &= \langle Jm'|e^{-i\alpha J_z}e^{-i\beta J_y}e^{-i\gamma J_z}|Jm\rangle \nonumber\\
%    &= e^{-im'\alpha}d^J_{m'm}e^{-im\gamma},
%\end{align}
The matrix element can be put into the form
\begin{align}
    B &= \sum_{m''_2}\sum_{m'}\langle Jm'|j_1m_1j_2m''_2\rangle D^{J\ast}_{mm'}(R^p_r)D^{j_2\ast}_{m''_2m_2}(R^p_v),
    \label{eqn:B_final}
\end{align}
where $\langle Jm'|j_1m_1j_2m''_2\rangle$ is the Clebsch-Gordan coefficient that connects the initial
and final configurations of angular momentum and the terms of the form $D^j_{mn}$ are Wigner D-Functions, with
$R^p_r$ being the passive rotation pointing towards the $K^\ast$ as measured in the helicity frame of the resonant meson
and $R^p_v$ being the passive rotation pointing towards the $\pi$ as measured in the helicity frame of the
$K^\ast$ vector-meson.   

%The expected intensity of efficiency corrected phase space distributions (determined
%by monte carlo) are weighted within the PWA using the  

%XXXXXXXXXXXXXXXXXX



Invariant mass distributions for $K^\ast\bar{K}$ with $J=0,1,2$ values determined by PWA can be seen
in Figures~\ref{fig:j0},~\ref{fig:j1} and \ref{fig:j2}, respectively. 
For each Figure~\ref{fig:j0} through~\ref{fig:j2}, the red line represents a total fit to the data.
For $J=0$ (Figure~\ref{fig:j0}) the green line represents the $\eta(1405)$, while 
blue is the $\eta(1475)$ and the yellow line is a third-degree polynomial background.
For $J=1$ (Figure~\ref{fig:j1}) the magenta line represents the $f_1(1420)$, the green line represents
$f_1(1510)$ and an exponential background is cyan.
For $J=2$ (Figure~\ref{fig:j2}) the green line represents the $f_2(1430)$, the blue line represents  
$f'_2(1525)$ while the background contribution is assumed as the tail of the $f_2(1260)$ shown in 
magenta.

\vspace{15pt}
\begin{figure}[h]
\begin{minipage}[t][2.5in][t]{.3\linewidth}
 { \centering
  \captionsetup{margin=0.5cm}
  \includegraphics[width=0.9\linewidth]{hMassAcc1Mod.png}
  \caption{$M(K^\ast\bar{K})$, $J=0$ from PWA.
  %\vspace{-5pt}
 }
  \label{fig:j0}}
\end{minipage}
\begin{minipage}[t][2.5in][t]{.3\linewidth}
 { \centering
  \captionsetup{margin=0.5cm}
  \includegraphics[width=0.9\linewidth]{hMassAcc2Mod.png}
  \caption{$M(K^\ast\bar{K})$, $J=1$ from PWA.
    %\vspace{-5pt} 
  }
  \label{fig:j1}}
\end{minipage}
\begin{minipage}[t][2.5in][t]{.3\linewidth}
 { \centering
  \captionsetup{margin=0.5cm}
  \includegraphics[width=0.9\linewidth]{hMassAcc3Mod.png}
  \caption{$M(K^\ast\bar{K})$, $J=2$ from PWA.
    %\vspace{-5pt}
  }
  \label{fig:j2}}
\end{minipage}
\vspace{-15pt}
\end{figure}
%\vspace{15pt}


The fit parameters, along with PDG values, for each of the extracted mesons shown in Figures~\ref{fig:j0}-\ref{fig:j2}
can be found in table~\ref{tab:j012}. 

%\vspace{-15pt}
\begin{table}[h]
\vspace{-15pt}
    \centering
    \begin{tabular}{c|c|c|c|c|c}
        $J$ & PID & PDG center & PDG width & Fit center & Fit width\\
            &     &   (MeV)    &   (MeV)   &     (MeV)  &    (MeV) \\
        \hline
        $0$ & $\eta(1405)$ & $1408.8\pm2.0$    & $50.1\pm2.6$  & $1406\pm2$ & $49.46\pm7.07$ \\
        $0$ & $\eta(1475)$ & $1475\pm4$    & $90\pm9$  & $1475\pm10$ & $105.8\pm2.24$\\
        $1$ & $f_1(1420)$  & $1426.3\pm0.9$  & $54.5\pm2.6$  & $1436\pm11$ & $48.40\pm4.17$\\
        $1$ & $f_1(1510)$  & $1518\pm5$    & $73\pm25$ & $1503\pm5$ & $71.78\pm12.76$\\
        $2$ & $f_2(1430)$  & ${\sim}1430$ & NA & $1438\pm4$ & $62.36\pm5.70$\\
        $2$ & $f_2(1525)$  & $1517.4\pm2.5$    & $86\pm5$  & $1538\pm4$ & $90.03\pm13.73$\\
    \end{tabular}
    \caption{Comparison of the Fit Parameters for Figures \ref{fig:j0}-\ref{fig:j2} 
With Respect to the PDG.
%\footnote{Fit width column is determined by $\Gamma=2.355\sigma$ for full width at half maximum.}.]
%{Comparison of the fit parameters for Figures \ref{fig:j0}, \ref{fig:j1}, and \ref{fig:j2} with res\
%pect to the PDG
%\footnote{Fit width column is determined by $\Gamma=2.355\sigma$ for full width at half maximum.}
\label{tab:j012}
}
\vspace{-5pt}
\end{table}
Of particular interest are the $\eta(1405)$ and $f_2(1430)$ states. The $\eta(1405)$ 
has not been seen in $\gamma\gamma$ collisions, pointing to a possible large gluonic
content, while the $f_2(1430)$ is a state that is omitted from the PDG Summary Tables
due to a paucity of data. Each of these potentially interesting states ($\eta(1405)$ and $f_2(1430)$) 
appear to show up rather strongly in our studies. Our evidence of $f_2(1430)$ will hopefully 
help more firmly establish that particle. 
%Further study of the $\eta(1405)$ will require
%that we obtain cross section information in order to compare to possible production mechanisms.
  

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\vspace{-10pt}
\subsubsection{Work on $\Xi^\ast$ states}
\vspace{-5pt}
%\vspace{55pt}

\begin{figure}[H]
\begin{minipage}[][1.5in][t]{.5\linewidth}
 { \centering
  \captionsetup{margin=0.4cm}
  \includegraphics[width=0.9\linewidth]{clVsCas.jpg}
  \caption{Mass of $\Lambda \pi^-$ versus confidence level.
%  \vspace{25pt}
  }
  \label{fig:clVsCas}}
\end{minipage}%                                                                                       
\begin{minipage}[][1.5in][t]{.5\linewidth}
 { \centering
  \captionsetup{margin=0.4cm}
  \includegraphics[width=0.9\linewidth]{cas.jpg}
  \caption{Invariant mass of $\Lambda \pi^-$ when confidence level is  $> 10^{-4}$. 
%    \vspace{25pt} 
  }
  \label{fig:cas}}
\end{minipage}
%\begin{minipage}[t][1.5in][t]{.3\linewidth}
% { \centering
%  \captionsetup{margin=0.4cm}
%  \includegraphics[width=0.9\linewidth]{casStar.jpg}
%  \caption{Invariant mass of $\Xi^- \pi^0$.
%    \vspace{25pt}
%  }
%  \label{fig:casStar}}
%\end{minipage}%                                                                                       
\vspace{35pt}
\end{figure}
\vspace{30pt}

%ASDFASDF


Graduate student Brandon Sumner is heading up the $\Xi^*$ analysis effort 
for the group. 
The analysis starts with a kinematic fit. The initial reaction of interest is
$\gamma p \rightarrow K^+ K^+ \pi^0 \Xi^-$, where $\Xi^- \rightarrow \Lambda \pi^-$. 
The $\Lambda$ and $\pi^0$ are constrained to the PDG masses within the kinematic fit, whereas
the $\Xi^-$ mass is not. 
A plot of the confidence level from kinematic fitting versus
the 
\begin{wrapfigure}[]{r}{0.45\textwidth}
\vspace{-20pt}
\begin{center}
\includegraphics[width=\linewidth]{Simfit.png}
\end{center}
\vspace{-20pt}
\caption{Simultaneous fit of Mass[$\Xi^*$] branches: $\Xi^{-*} \rightarrow K^- \Lambda$ (top panel)
and $\Xi^{-*} \rightarrow \Xi^- \pi^0$ (bottom panel).
}
{\label{fig:simFit}}
\end{wrapfigure}
invariant mass of $\Lambda \pi^-$ is shown in Figure~\ref{fig:clVsCas}. 
Mr. Sumner performed a study to determine the confidence level that minimizes the 
fractional uncertainty of extracted $\Xi(1520)$ states, and found it best to accept all events with
a confidence level greater than $10^{-4}$.
Upon a confidence level of $10^{-4}$, an invariant mass of $\Lambda \pi^-$
given in Figure~\ref{fig:cas} is determined, where the peak at 1.320 GeV is the ground state $\Xi$. 
The $\Xi$ is then defined as those events with invariant mass $\Lambda \pi^-$ between 1.31 and 1.34 GeV. 




In addition to exploring the branch $\Xi^{-*} \rightarrow \Xi^- \pi^0$, we investigate the branch 
$\Xi^{-*} \rightarrow K^- \Lambda$. 
For the case of $\Xi^{-*} \rightarrow K^- \Lambda$, the $\Lambda$
is not constrained and a confidence level study is performed on the $\Xi(1820)$ in a similar manner 
as described for
the $\Xi^{-*} \rightarrow \Xi^- \pi^0$ branch using $\Xi(1530)$.  
%The resulting invariant mass of the $\Xi^- \pi^0$ is shown in figure~\ref{fig:casStar}
%In addition to the branch of $\Xi^*$ going $\Xi \pi$ we have begun to explore
%the $\Xi^* \rightarrow K \Lambda$ branch. 

%The invariant mass
%of $K^- \Lambda$ from the reaction $\gamma p \rightarrow K^+ K^+ K^- \Lambda$ 
%is shown in the top panel of figure~\ref{fig:simFit} and the invariant mass of 
%$\Xi^- \pi^0$ shown on the bottom panel.



A simultaneous fit was performed on $\Xi^{-*}$ states utilizing 2018 data for both branches described above. 
Figure~\ref{fig:simFit} shows the invariant mass of $K^- \Lambda$ from the reaction $\gamma p \rightarrow K^+ K^+ K^- \Lambda$
in the top panel, while the bottom panel shows the invariant mass of $\Xi^- \pi^0$ from the reaction
$\gamma p \rightarrow K^+ K^+ \Xi^- \pi^0$. The fit consists of four $\Xi^{-*}$ states 
($\Xi(1530), \Xi(1620)$, $\Xi(1690)$, $\Xi(1820)$) along with polynomial backgrounds. 
The simultaneous fit constrained the
%The simultaneous fit constrained the 
center and widths of the $\Xi^*$ states to be identical for 
both top and bottom panels of Figure~\ref{fig:simFit}, but allowed the 
amplitudes to vary independently. For both branches within the simultaneous fit, the background was
described by a third degree polynomial with parameters being independent between the branches.
Within Figure~\ref{fig:simFit}, solid lines represent the full fit, dashed-dotted lines
are the polynomial backgrounds with solid colors representing the $\Xi^*$ states, where blue is $\Xi(1530)$, turquoise represents $\Xi(1620)$,
green is the $\Xi(1690)$ and red is the $\Xi(1820)$.


The $\Xi(1530)$ is a four-star state, $\Xi(1690)$ and $\Xi(1820)$ are three-star, and the $\Xi(1620)$ is a one-star
that is omitted from the summary table. Although the $\Xi(1690)$ has a three-star designation, the PDG still lists the $J^P$ as unknown with
$J^P = ?^?$ with an unknown branching ratio  $\Gamma(K^- \Lambda)/\Gamma(\Xi^- \pi^0)$ that we plan to measure.  


%two $\Xi^*$ states: $\Xi(1690)$ and $\Xi(1820)$ with a smooth polynomial
%background shown as a dash-dotted line. The $\Xi(1820)$ is a three-star state, but the $\Xi(1950)$
%lacks a PDG designation for the $J^P$ quantum numbers, whereas the $\Xi(1820)$ has
%a PDG designation $J^P=\frac{3}{2}^-$. This is another example where a PWA will be useful
%in helping assign quantum numbers to excited $\Xi$ states. Moreover, since the $\Xi(1690)$
%is common to both branches shown here (figures~\ref{fig:csCas2} and \ref{fig:cas3}), a simultaneous fit
%of those branches will be used to uncover the 
%$\Gamma(\Xi^- \pi^0)/\Gamma(K^- \Lambda)$ 
%branching ratio for the first time (once we have
%calculated the detection efficiencies for the $K^- \Lambda$).


\begin{wrapfigure}[]{r}{0.4\textwidth}
\vspace{-10pt}
\begin{center}
%\includegraphics[width=\linewidth]{CSupdatedv2.png}
\includegraphics[width=\linewidth]{CrossSectionPic.png}
\end{center}
\vspace{-20pt}
\caption{Preliminary cross section of $\Xi^-(1530)$.
\newline
\vspace{-8pt}
}
{\label{fig:csCas}}
\end{wrapfigure}
Detector simulation studies have been performed for the $\Xi(1530)$ state to allow for detector efficiencies to
be calculated, but there remains work to be done towards improving the event generator.
The generator employed in determining the detection efficiencies is a phase space generator that
considers the initial part of the reaction $\gamma p \rightarrow K^+ K^+ \Xi^-$ as a 
one-step process. However, due to OZI suppression, a two step process is far more likely.  
Brandon Sumner has nearly completed work on a generator that better matches the reaction of 
interest: $\gamma p \rightarrow K^+ Y$, where $Y$ is a very wide high-mass hyperon that decays
$Y \rightarrow K^+ \Xi^{-*}$. 
Preliminary $\gamma p \rightarrow K^+ K^+ \Xi(1530)$ cross sections in term of 
incident photon energy are shown in Figure~\ref{fig:csCas}. 

%Preliminary cross section of
%background subtracted $\gamma p \rightarrow K^+ K^+ \Xi \pi^0$ is shown in figure~\ref{fig:csCas2}, 
%with the solid black line representing a fit to three $\Xi^*$ states: 
%$\Xi(1530)$
%(center = 1536(1) MeV, width = 30(4) MeV)
%, $\Xi(1620)$ 
%(center = 1606(4) MeV, width = 15(7)) 
%and $\Xi(1690)$. 
%(center = 1687(6) MeV, width = 10(6))
%While the $\Xi(1530)$ is a well known four-star 
%resonance, the remaining two are less well understood. The neither the $\Xi(1620)$ (one-star) nor
%the $\Xi(1690)$ (three-star) has confirmed $J^P$ values assigned in the PDG. A PWA will have
%to be performed in order for our data to contribute to the discussion of $J^P$ assignments for
%these states.


%\vspace{-10pt}
%\subsubsection{Quality factors}
%For each of our reactions, we are exploring the use of quality factors for the reduction of 
%background. For each event, a quality factor is assigned by taking the nearest events in phase space
%and determining the probability (through a fitting procedure) that the event is signal. 
%\vspace{-5pt}

\vspace{-10pt}
\subsection{Publications during the current grant}
\vspace{-5pt}

Several papers have been published during the current grant period
~\cite{CLAS:2021udy,CLAS:2021osv,Hayward:2021psm,CLAS:2020ngl,PhysRevC.103.L022201,
CLAS:2020vpr,ADHIKARI2021164807,CLAS:2020rdz,PhysRevC.101.065206,CLAS:2020mom,
PhysRevC.100.052201,PhysRevLett.123.072001,CLAS:2018mmb}.
Of those papers, we had direct involvement in writing the text related to the triplet polarimeter 
for the GlueX beamline and detector paper~\cite{ADHIKARI2021164807} and for providing the 
polarization values used in the determination of beam asymmetries for the photoproduction 
from the proton for
$\eta$, $\eta^{\prime}$\cite{PhysRevC.100.052201};
$K^+ \Sigma^0$ \cite{PhysRevC.101.065206} and 
$\pi^- \Delta^{++}$\cite{PhysRevC.103.L022201}. 
We provided the initial event generator used
for background studies for the early analysis of $J\psi$ that eventually led to 
publication~\cite{PhysRevLett.123.072001}. The $\pi^- \Delta^{++}$\cite{PhysRevC.103.L022201}
paper utilized a method developed by the PI for the determination of beam 
asymmetries~\cite{PhysRevC.88.065203}
and the PI had discussions with the person that led the GlueX analysis effort (J. Zarling) regarding the
implementation of that method. The PI was heavily involved with the FROST group prior to this current grant
and two of the papers \cite{CLAS:2021udy,CLAS:2018mmb} were from those efforts. The remaining papers had
involvement from group members that was less substantial than described above.   
 
    

%GlueX NIM \cite{ADHIKARI2021164807}. 
%
%$\mathrm{\ensuremath{\Sigma}}$ for the photoproduction of $\ensuremath{\eta}$ 
%and ${\ensuremath{\eta}}^{\ensuremath{'}}$ mesons at 
%${E}_{\ensuremath{\gamma}}=8.8\phantom{\rule{0.28em}{0ex}}\mathrm{GeV}$ 
%\cite{PhysRevC.100.052201}, 
%
%First Measurement of Near-Threshold $J/\ensuremath{\psi}$ Exclusive Photoproduction off the Proton
%\cite{PhysRevLett.123.072001},
%
%Measurement of the photon beam asymmetry in 
%$P{\ensuremath{\gamma}}\cite{PhysRevC.101.0
%$p\ensuremath{\rightarrow}{K}^{+}{\mathrm{\ensuremath{\Sigma}}}^{0}$ 
%at ${E}_{\ensuremath{\gamma}}=8.5$ GeV
%\cite{PhysRevC.101.065206},
%
%Measurement of beam asymmetry for ${\ensuremath{\pi}}^{\ensuremath{-}}{\mathrm{\ensuremath{\Delta}}}^{++}$ 
%photoproduction on the proton at ${E}_{\ensuremath{\gamma}}=8.5\phantom{\rule{0.16em}{0ex}}\mathrm{GeV}$
%\cite{PhysRevC.103.L022201}
%paper used method I developed \cite{PhysRevC.88.065203} for the analysis of beam asymmetry measurements 
%and I had discussions with the person that led the GlueX analysis effort (J. Zarling) regarding the
%implementation.

